Author's response to reviews

Title: Validity of subjective versus objective quality of life assessment in people with schizophrenia

Authors:
Karen P Hayhurst (karen.hayhurst@manchester.ac.uk)
Jennifer A Massie (Jenny.Massie@mhsc.nhs.uk)
Graham Dunn (graham.dunn@manchester.ac.uk)
Shôn W Lewis (shon.lewis@manchester.ac.uk)
Richard J Drake (richard.drake@manchester.ac.uk)

Version: 2
Date: 13 August 2014

Author's response to reviews:
N/A - only changes are to add line numbers to manuscript